Author's response to reviews

Title: Effects of self-monitoring of glucose in non-insulin treated patients with type 2 diabetes: Design of the IN CONTROL-trial

Authors:

Uriëll L Malanda (u.malanda@vumc.nl)
Sandra DM Bot (s.bot@vumc.nl)
Piet J Kostense (pi.kostense@vumc.nl)
Frank J Snoek (fj.snoek@vumc.nl)
Jacqueline M Dekker (jm.dekker@vumc.nl)
Giel Nijpels (g.nijpels@vumc.nl)

Version: 2 Date: 10 March 2009

Author's response to reviews:

Amsterdam March 10, 2009

Dear Editor,

Hereby we would like to submit our manuscript entitled ‘Effects of self-monitoring of glucose in non-insulin treated patients with type 2 diabetes: Design of the IN CONTROL-trial’.

This article presents a detailed description of an RCT, designed with the aim to investigate the effects of self-monitoring of glucose on diabetes specific emotional distress and self-efficacy in patients with type 2 diabetes who are not using insulin compared to usual care. Publication of this article will provide researchers, health care providers and policy makers the opportunity to critically review the methodological quality, the background theory and the practical issues of the RCT.

The paper is not under consideration for publication elsewhere and no parts of the paper have been published previously. All authors were involved in drafting the article and revising it critically. The final version of the manuscript is approved by all authors. There are no conflicts of interest.

We hope you will consider publication of this study in BMC Family Health

Sincerely,

Uriëll Malanda
Corresponding author